

A mating-induced reproductive gene promotes Anopheles tolerance to Plasmodium falciparum infection

Perrine Marcenac, W. Robert Shaw, Evdoxia Kakani, Sara Mitchell, Adam South, Kristine Werling, Eryney Marrogi, Daniel Abernathy, Rakiswendé Serge Yerbanga, Roch Dabiré, et al.

▶ To cite this version:

Perrine Marcenac, W. Robert Shaw, Evdoxia Kakani, Sara Mitchell, Adam South, et al.. A mating-induced reproductive gene promotes Anopheles tolerance to Plasmodium falciparum infection. PLoS Pathogens, 2020, 16 (12), pp.e1008908. 10.1371/journal.ppat.1008908. hal-03450556

HAL Id: hal-03450556 https://hal.science/hal-03450556

Submitted on 26 Nov 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



GOPEN ACCESS

Citation: Marcenac P, Shaw WR, Kakani EG, Mitchell SN, South A, Werling K, et al. (2020) A mating-induced reproductive gene promotes *Anopheles* tolerance to *Plasmodium falciparum* infection. PLoS Pathog 16(12): e1008908. <u>https://</u> doi.org/10.1371/journal.ppat.1008908

Editor: Kenneth D. Vernick, Institut Pasteur, FRANCE

Received: August 24, 2020

Accepted: November 11, 2020

Published: December 21, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1008908

Copyright: © 2020 Marcenac et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data and code files are available from the Harvard Dataverse repository: https://doi.org/10.7910/DVN/GU8DOA.

RESEARCH ARTICLE

A mating-induced reproductive gene promotes *Anopheles* tolerance to *Plasmodium falciparum* infection

Perrine Marcenac¹, W. Robert Shaw¹, Evdoxia G. Kakani^{1¤a}, Sara N. Mitchell^{1¤a}, Adam South^{1¤b}, Kristine Werling¹, Eryney Marrogi¹, Daniel G. Abernathy¹, Rakiswendé Serge Yerbanga², Roch K. Dabiré², Abdoulaye Diabaté², Thierry Lefèvre^{2,3}, Flaminia Catteruccia^{1*}

1 Department of Immunology and Infectious Diseases, Harvard T. H. Chan School of Public Health, Boston, Massachusetts, United States of America, 2 Institut de Recherche en Sciences de la Santé/Centre Muraz, Bobo-Dioulasso, Burkina Faso, 3 MIVEGEC, IRD, CNRS, University of Montpellier, Montpellier, France

¤a Current address: Verily Life Sciences, South San Francisco, California, United States of America
¤b Current address: Department of Infectious Disease and Global Health, Cummings School of Veterinary
Medicine at Tufts University, Grafton, Massachusetts, United States of America
* fcatter@hsph.harvard.edu

Abstract

Anopheles mosquitoes have transmitted Plasmodium parasites for millions of years, yet it remains unclear whether they suffer fitness costs to infection. Here we report that the fecundity of virgin and mated females of two important vectors—Anopheles gambiae and Anopheles stephensi-is not affected by infection with Plasmodium falciparum, demonstrating that these human malaria parasites do not inflict this reproductive cost on their natural mosquito hosts. Additionally, parasite development is not impacted by mating status. However, in field studies using different P. falciparum isolates in Anopheles coluzzii, we find that Mating-Induced Stimulator of Oogenesis (MISO), a female reproductive gene strongly induced after mating by the sexual transfer of the steroid hormone 20-hydroxyecdysone (20E), protects females from incurring fecundity costs to infection. MISO-silenced females produce fewer eggs as they become increasingly infected with P. falciparum, while parasite development is not impacted by this gene silencing. Interestingly, previous work had shown that sexual transfer of 20E has specifically evolved in Cellia species of the Anopheles genus, driving the co-adaptation of MISO. Our data therefore suggest that evolution of male-female sexual interactions may have promoted Anopheles tolerance to P. falciparum infection in the Cellia subgenus, which comprises the most important malaria vectors.

Author summary

Plasmodium falciparum, the deadliest form of human malaria, is transmitted when female *Anopheles* mosquitoes bite people and take a blood meal in order to develop eggs. To date, it is still poorly understood whether *Anopheles* mosquitoes that get infected with *P. falciparum* suffer fitness costs. Here, we find that the number of eggs produced by *Anopheles*

Funding: F.C. is funded by a Faculty Research Scholar Award by the Howard Hughes Medical Institute (HHMI) and the Bill & Melinda Gates Foundation (BMGF) (Grant ID: OPP1158190, www. hhmi.org), and by the National Institutes of Health (NIH) (R01 AI124165, R01 AI104956, www.nih. gov). P.M. was supported by the National Science Foundation Graduate Research Fellowship Program (Grant No. DGE1144152, www.nsf.gov), the Barry R. and Irene Tilenius Bloom Fellowship at the Harvard T. H. Chan School of Public Health (www.hsph.harvard.edu), and by a Merit/Termtime Research Fellowship from the Harvard University Graduate School of Arts and Sciences (www.gsas.harvard.edu). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

gambiae and *Anopheles stephensi* females is not affected by *P. falciparum* infection, and that the mating status of the mosquitoes does not impact the parasite. However, in field experiments infecting a related species, *Anopheles coluzzii*, with *P. falciparum* using blood from donors in Burkina Faso, we find that interfering with the expression of a gene normally triggered by the sexual transfer of the steroid hormone 20-hydroxyecdysone induces increasing costs to egg development as females become more infected with *P. falciparum*, with no impacts on the parasite. The results of our study suggest that pathways triggered by mating may help *Anopheles* prevent reproductive costs associated with *P. falciparum* infection, providing new insights into evolutionary strategies adopted by anophelines in the face of a longstanding association with *Plasmodium* parasites.

Introduction

Parasites of the genus *Plasmodium* are the causative agents of malaria, a disease that infects hundreds of millions and kills over 400,000 people, mostly children, every year [1]. *Plasmo-dium* transmission occurs via the infectious bite of mosquitoes of the *Anopheles* genus, which comprises three major subgenera whose last common ancestor dates to 100 million years ago [2,3]: *Cellia* found in Africa, the Indian sub-continent, South East Asia, and Oceania; *Nyssor-hynchus* located in Central and South America; and *Anopheles* distributed from North America to Asia [4]. *Cellia* species from the African continent have been vectors of *Plasmodium falciparum*, the deadliest malaria parasite, for the last 10,000–50,000 years [5–8], and have transmitted ancestral *Plasmodium* species for millions of years [9]. Given their long-term association with *P. falciparum* and its ancestral forms, an intriguing question is whether these vectors suffer fitness costs to infection.

Two major fitness costs possibly associated with infection are reduced longevity and impaired reproductive output. A meta-analysis on studies examining the effects of infection on mosquito survival found no costs to longevity when mosquitoes were infected with P. fal*ciparum*, while infections with rodent malaria parasites reduced female lifespan [10]. Moreover, rodent *Plasmodium* species also reduced the ability of females to develop eggs [11-16]. We recently began directly addressing whether P. falciparum induces reproductive costs in Anopheles gambiae, its predominant natural Cellia vector in sub-Saharan Africa [17]. Using virgin females, we unexpectedly found that egg numbers were positively correlated with P. falciparum infection intensity, so that females that produced more eggs in their ovaries harbored greater parasite numbers in their midgut. This correlation was shown to be mediated by 20-hydroxyecdysone (20E)—a steroid hormone produced by the female after blood feeding that plays an essential role in egg development [17-19]—as the positive link between parasite development and oogenesis was abolished by silencing either component of the 20E heterodimer nuclear receptor (ecdysone receptor EcR and ultraspiracle USP) [17]. Moreover, we and others have found that parasites can utilize lipids carried by the mosquito lipid transporter lipophorin for their own growth [17,20,21], but they do so without impairing egg development [17]. Although these results suggest a non-competitive *P. falciparum* developmental strategy tailored to An. gambiae reproductive biology [17], the study used virgin females and therefore did not determine the full costs potentially inflicted by parasites in reproductively active, mated females.

Mating profoundly reshapes female physiology, including initiating sperm storage and preservation in the spermatheca, a costly process that is essential for fertilization [22–24]. As a possible consequence of this increased investment in reproductive functions, mated

females may be less equipped to face *P. falciparum* infection, shifting the balance towards a competitive vector-parasite association. This issue is potentially exacerbated by the fact that female *An. gambiae* are monandrous, i.e., they mate a single time in their lives and therefore must preserve sperm viability over their entire lifespan [25,26]. Interestingly, monandry in this species is enforced by the same ecdysteroid hormone 20E, in this case produced by males in their accessory glands and transferred to the female atrium (uterus) during mating [27]. Besides enforcing monandry, male 20E transfer is responsible for an array of post-mating responses critical for female reproduction, including improved fertility, triggering of egg laying, and induction of large transcriptional changes in the female atrium [26–29]. Among the genes strongly upregulated by sexual transfer of 20E is the *Mating-Induced Stimulator of Oogenesis (MISO*), whose expression is specifically induced in the atrium where this female reproductive protein interacts directly or indirectly with 20E, favoring its function [28].

Interestingly, phylogenetically-based evolutionary analyses have revealed that sexual transfer of 20E is an acquired trait that has evolved specifically in *Cellia* [29] where levels of this male steroid hormone are highly variable between species, and it is largely absent in males from the *Nyssorhynchus* and *Anopheles* subgenera [29,30]. In turn, evolution of this male reproductive trait has induced reciprocal adaptations in female reproductive genes, including *MISO*, profoundly shaping female post-mating biology [29].

In an effort to better understand the interaction of *P. falciparum* parasites with their Anopheles vectors and gain insights into possible reproductive fitness costs associated with infection, we examined whether P. falciparum impairs an important component of female reproductive fitness—egg development after the first blood meal—in mated females using An. gambiae as well as Anopheles stephensi, a Cellia vector from the Indian sub-continent. We show that the fecundity of both anophelines is not impacted by infection regardless of mating status, and that *P. falciparum* infection intensity and prevalence are not affected by mating. However, in field studies using natural P. falciparum isolates, we uncover fitness costs to infection when we silence the 20E-responsive gene MISO, as MISO-depleted females produce fewer eggs when they become increasingly infected, while parasite numbers are unaffected. Therefore, this mating-induced gene supports the reproductive fitness of P. falciparum-infected Anopheles females not by limiting parasite infection, but rather by reducing fitness costs associated with it, an evolutionary strategy known as tolerance [31–33]. All together, these studies demonstrate that reproductively active females of Cellia species do not suffer costs to their fecundity when infected with P. falciparum, and they reveal a mating-induced factor critical to maintaining mosquito tolerance to Plasmodium infection.

Results

An. gambiae and An. stephensi females do not suffer fecundity costs to *P. falciparum* infection regardless of their mating status

We analyzed the effects of infection on egg development in both *An. gambiae* (G3 strain) and *An. stephensi* (SDA-500 strain) mated females. After mating, females from both species were fed on a blood meal containing *P. falciparum* (NF54) parasites or a heat-inactivated, non-infectious blood meal, using virgins as controls. Forty-eight hours (h) post-infectious blood feeding (pIBF), the number of eggs developed per mosquito was counted, while mated status was confirmed by detecting successful mating plug transfer prior to blood feeding (see <u>Materials and Methods</u>). In both species, virgin and mated females produced the same number of eggs whether the blood meal was infectious or non-infectious (Fig 1A and 1B, see S1 Table for all summary statistics), supporting a non-competitive interaction with *P. falciparum* as previously reported for virgin *An. gambiae* [17]. Of note, the number of eggs developed by mated





Fig 1. P. falciparum infection does not affect the fecundity of An. gambiae and An. stephensi females. P. falciparum infection does not affect the number of eggs developed by virgin and mated (A) An. gambiae females (infection: likelihood ratio test (LRT) $X_{1}^{2} = 0.76$, p = 0.38; mating*infection: LRT $X^2_1 = 3.1$, p = 0.08) or (B) An. stephensi females (infection: LRT $X^2_1 = 0.007$, p = 0.94; mating*infection: LRT $X_1^2 = 0.99$, p = 0.32). In An. stephensi, mating increased the likelihood of egg development (LRT $X_1^2 = 16.26$, p < 0.001) and the number of eggs developed (mating: LRT $X_1^2 = 20.22$, p < 0.001). The data from this graph are from three replicates for each Anopheles species. Pairwise comparisons (letters) are computed via estimated marginal means. Lines represent medians, and n represents sample size. Open circles represent virgin females, closed circles represent mated females, and light and dark colors correspond to uninfected and infected females, respectively. ns = not statistically significant.

https://doi.org/10.1371/journal.ppat.1008908.g001

An. gambiae females was similar to virgins, as reported in some studies [30] but not in others [28] (Fig 1A, S1 Table). In An. stephensi, mating not only increased the likelihood of egg development but also boosted the number of eggs developed (Fig 1B, S1 Table), consistent with previous studies [29,30].

Mating does not affect laboratory P. falciparum infections in An. gambiae and An. stephensi

In a subset of the virgin and mated An. gambiae and An. stephensi females used in the fecundity experiments detailed above, parasite intensity and prevalence of infection were determined by counting midgut oocysts 7 days pIBF. Interestingly, we detected no effects of mating on *P. falciparum* infection in *An. gambiae* (Fig 2A), as the same proportion of females became infected (infection prevalence, pie charts), and females developed the same number of oocysts (infection intensity) regardless of mating status (S1 Table). Similar results were obtained using An. stephensi, where again we observed no difference in prevalence and intensity of infection linked to mating (Fig 2B, S1 Table). In our hands, therefore, mating has no effect on P. falciparum infection in two different Anopheles species, in contrast to a recent study reporting that mating increases the susceptibility of Anopheles coluzzii females, a sibling species of An. gam*biae*, to *P. falciparum* [34].

The mating-induced gene MISO mediates mosquito tolerance to P. falciparum in field infections

Our previous studies in virgin An. gambiae had shown that the steroid hormone 20E produced by females after blood feeding regulates the positive interaction between parasite and egg



Fig 2. Mating status does not impact *P. falciparum* **laboratory infections in** *An. gambiae* **and** *An. stephensi.* Mating status does not affect prevalence (P, indicated with pie charts below the graph) or intensity of infection in (A) *An. gambiae* females (prevalence: LRT $X_1^2 = 1.08$, p = 0.30; intensity: LRT $X_1^2 = 0$, p = 1.00) or (B) *An. stephensi* females (prevalence: LRT $X_1^2 = 0.923$; intensity: LRT $X_1^2 = 1.47$, p = 0.23). The data from this graph are from three replicates for each *Anopheles* species. Pairwise comparisons are computed via estimated marginal means. Lines represent medians, and n represents all data points including females that failed to develop occysts. Females that developed no occysts are omitted from the dot plots to represent intensity of infection. Open circles represent virgin females, closed circles represent mated females. ns = not statistically significant.

https://doi.org/10.1371/journal.ppat.1008908.g002

numbers [17]. Therefore, we set out to investigate whether mating-induced processes regulated by sexual transfer of the same steroid hormone 20E might be critical to the relationship between oogenesis and P. falciparum development in mated females. To this aim, we performed infections after silencing MISO, a female reproductive gene potently induced by male 20E in the female atrium after mating and that forms a complex with this steroid hormone, influencing timely lipid accumulation during oogenesis [28]. We reasoned that depleting MISO would be an effective strategy to partially disrupt male 20E-induced processes important for egg development and determine their impact during the course of *P. falciparum* infection. To gain a comprehensive picture of natural Anopheles-Plasmodium associations, infections were carried out in Burkina Faso using P. falciparum-infected blood collected from 5 donors with a range of gametocytemia values (S2 Table) and female mosquitoes from an An. coluzzii strain recently colonized from this region. After MISO depletion via RNAi injections of dsRNA targeting MISO (dsMISO), we allowed these females (or dsLacZ-injected controls, dsControl) to mate prior to being fed a P. falciparum-infected blood meal. We then deprived them of the opportunity to lay eggs (females retain eggs in these conditions, see Materials and Methods) in order to collect paired egg-oocyst data for each individual after dissecting females 7 days pIBF. The occurrence of mating was confirmed by detecting the presence of sperm in the spermatheca at the time of dissection.

When we analyzed oogenesis across different gametocytemia values, we found that gametocytemia alone had no effect on the number of eggs developed by *An. coluzzii* females (S1 Table). Egg development in control females appeared to be more dependent on the random blood donor rather than on gametocytemia (Fig 3A). In contrast, we found that the interaction of gene silencing and gametocytemia significantly affected the number of eggs developed (S1 Table), whereby ds*MISO* females produced significantly fewer eggs at high gametocytemia infections relative to lower gametocytemia infections (Fig 3A). These data suggest that highly infectious *P. falciparum* blood meals reduce the reproductive fitness of females when this 20E-induced reproductive gene is depleted.

In both ds*MISO* and control groups, high gametocytemia in the donor blood resulted in higher oocyst loads, while infection prevalence was close to 70% or greater in all conditions and was not affected by gametocytemia (Fig 3B, S1 and S2 Tables). Interestingly, no differences in either infection prevalence or intensity were observed in *MISO*-silenced versus control females, demonstrating that this reproductive gene does not impact parasite development (Fig 3B, S1 Table).

Additionally, when we examined the relationship between egg and oocyst numbers across all *An. coluzzii* infections by using paired egg-oocyst data, we found that in ds*MISO*-injected females egg numbers decreased with increasing oocyst loads, while they were not affected by infection in ds*Control* females (**Fig 3C**, **S1 Table**). Consistent with these findings, when analyzing the probability of egg development across the observed range of oocyst intensities, we found that while control individuals had a greater chance of developing eggs as the oocyst burden increased (**Fig 3D**), ds*MISO* females had a markedly reduced probability of egg development at increasingly high parasite loads (**Fig 3D**; **S1 Table**) and were 2.3% less likely to develop eggs for every additional oocyst they developed (Unit odds ratio = 0.977).

Combined, these data reveal that in the face of high intensity *P. falciparum* infections, MISO helps preserve reproductive fitness—as assessed by female fecundity—without affecting parasite survival, thereby acting as a key factor mediating *Anopheles* tolerance to *P. falciparum*.

Discussion

In this study, we demonstrate that *P. falciparum* infections do not affect the fecundity of two important anopheline vectors—An. gambiae and An. stephensi—regardless of their mating status. Moreover, in An. coluzzii infections using natural P. falciparum isolates, we found that the number of eggs developed was not impaired by infections with increasing intensity, confirming previous data suggesting that P. falciparum has adopted a non-competitive evolutionary strategy in its natural mosquito vectors. Additionally, we found that females that have a higher likelihood of developing eggs harbor increasing oocyst loads, which expand on our previous findings in virgin An. gambiae where we found that egg and parasite numbers are positively correlated [17]. It is worth noting that past studies that showed costs of infection imposed by *Plasmodium* spp. were performed using rodent malaria parasites that are not naturally transmitted by anthropophilic Anopheles species [11–16]. The positive relationship between egg development and parasite numbers in this study and in our previous work [17] may be explained by the theory that organisms must balance resources between somatic, reproductive, and immune processes [35], whereby An. coluzzii females that devote more resources to egg development may have a suppressed immune response, enabling higher P. falciparum oocyst loads. Alternatively, processes triggered by blood feeding and regulating egg development post-mating may affect mosquito physiology in ways that also affect the probability of parasite survival in the midgut, while not being directly aimed at the parasite. A caveat of our study is that we only assessed the effects of parasite infection on the number of eggs developed after a first blood meal. Future studies will also be needed to address whether parasites may affect other components of mosquito fitness, including survival, fertility, and egg development over multiple gonotrophic cycles.



Fig 3. MISO enforces tolerance to P. falciparum in field infections with gametocyte carriers. (A) The number of eggs developed by An. coluzzii females silenced for MISO (dsMISO) is negatively impacted by P. falciparum infections with high gametocytemias, while this effect is not observed in dsControl mosquitoes (treatment* gametocytemia: LRT X_3^2 = 16.15, p = 0.001). Four gametocytemias were tested (see S2 Table), with each dot representing paired egggametocytemia data for one female. (B) Both dsControl and dsMISO females harbor more oocysts when they are fed with higher gametocytemia blood meals (gametocytemia: LRT $X_3^2 = 20.36$, p < 0.001), with no impact of *MISO* silencing (treatment: LRT $X_1^2 = 0.85$, p = 0.36; treatment*gametocytemia: LRT $X_3^2 = 0.85$, p = 0.84). P. falciparum prevalence (P, pie charts) is not affected by gametocytemia or MISO silencing (gametocytemia: LRT $X_3^2 = 5.78$, p = 0.12; treatment: LRT $X_{1}^{2} = 0.60$, p = 0.44; treatment*gametocytemia: LRT $X_{3}^{2} = 4.63$, p = 0.20). Each dot represents paired oocyst-gametocytemia data for one female. For both A and B, data are represented as box and whisker plots with minimum and maximum values. Pairwise comparisons (letters) are computed via estimated marginal means. In B, females that developed no oocysts are omitted from the dot plots to represent intensity of infection. (C) MISO silencing breaks the interaction between egg and oocyst development, with higher oocyst intensities leading to fewer numbers of eggs developed (treatment*oocyst number: LRT $X_1^2 = 7.77$, p = 0.005). Each dot represents paired egg-oocyst data for one female. One data point for dsControl (178, 88) is excluded from the graph to simplify data visualization. Dotted regression lines represent data trends only. tx = treatment, n = number, * = interaction. (D) While in control females the probability of egg development increases as oocyst intensity increases, MISO-silenced females have a lower likelihood of egg development as infection burden increases (treatment*oocyst number: LRT $X_1^2 = 7.18$, p = 0.007). Curves represent the probability of egg development given x number of oocysts, with shading indicating 95% confidence intervals. In all panels, n represents sample size. All panels in this figure come from data collected from three independent replicates.

https://doi.org/10.1371/journal.ppat.1008908.g003

Moreover, we show that mating in *An. gambiae* and *An. stephensi* does not affect the intensity or the prevalence of *P. falciparum* infection. These data are in contrast to a recent report showing that mated *An. coluzzii* females harbor increased parasite burdens relative to virgins [34]. While we cannot explain the reasons behind this discrepancy, our findings were reproducibly obtained using two different *Anopheles* species.

The observation of no costs to fecundity induced by infection might be explained by considering that the interactions of P. falciparum and its ancestral lineages with Cellia species have been significant on an evolutionary scale. Indeed, human Plasmodia originated from ape populations in regions of Africa where these anophelines have thrived for millions of years (reviewed in [9]). Co-adaptations between mosquitoes and parasites may have arisen due to this longstanding association, favoring the evolution of tolerance mechanisms in the mosquito to limit costs of infections and of immune avoidance strategies in the parasite to avoid the induction of host resistance. A powerful example of the latter is provided by the evolution of the parasite gene *Pfs47* which mediates *P. falciparum* immune evasion in the mosquito [36]. This gene has undergone selection due to pressures imposed by the mosquito immune system in different Anopheles species, leading to distinct, geographically structured haplotypes [36– 38]. Consistent with this scenario of mosquito-parasite co-evolution, we found evidence of a tolerance-mediating factor in An. coluzzii—one of the most important vectors of P. falciparum in sub-Saharan Africa. Specifically, in field studies we determined that the female protein MISO helps maintain mosquito fitness in the face of infection. Reproductive costs inflicted by parasites when MISO was depleted increased with higher infection intensities, while this depletion did not affect the parasite itself. Tolerance is a defense strategy whereby an organism reduces the fitness costs associated with an infection without impacting pathogen burden [31-33], and as such our results identify MISO as critical to An. coluzzii tolerance to P. falciparum. MISO is strongly induced in the female atrium by sexual transfer of 20E, which among mosquitoes has specifically evolved in the Cellia subgenus, with African species transferring higher titers of this steroid hormone [29]. Intriguingly, the evolution of MISO appears to have been driven by male 20E transfer, as MISO orthologues in species with the highest levels of male 20E show a greater degree of similarity than expected based solely on phylogenetic relationships [29]. When combined, our data suggest that the evolution of MISO may have promoted Anopheles tolerance to P. falciparum, thus contributing to the non-competitive strategy between these two organisms [17]. Tolerance strategies have previously been described in insects, including Aedes mosquitoes that transmit dengue and chikungunya [39]; however, to our knowledge this is the first demonstration of tolerance in an Anopheles vector of human malaria.

How could MISO affect female reproductive fitness during infection? Our previous work has shown that MISO ensures timely accumulation of lipids in developing eggs [28]. This protein forms a complex with male 20E [28], most likely in the atrium and ampullae, the tissues which connect the atrium to the ovaries. It is possible, then, that this complex may ensure the local release of sexually transferred 20E to preserve egg development in the face of midgut stress signals exacerbated by high infection loads. This model is compatible with known roles of steroid hormones in limiting damage induced by, for instance, oxidative stress [40–42]. Alternatively, the MISO-20E interaction may permanently 'prime' the ovaries to be competent for the correct accumulation of nutrients in mated females after blood feeding, a process known to be regulated by synthesis of the same hormone 20E in the female fat body. More recently, we have demonstrated that when blood meal-derived resources accumulate in the midgut rather than being delivered to the ovaries, as for instance when we impair egg development by silencing the 20E nuclear receptor EcR/USP, parasites are capable of utilizing those resources for growth [17]. It is therefore possible that when *MISO* is silenced, mosquito

resources that cannot readily be taken up by the ovaries may instead be utilized by parasites in some way, leading to a decrease in egg numbers that becomes more significant with higher parasite loads. Other explanations, including competition for resources due to sperm preservation in mated females, cannot be ruled out, and studying the cascades of events induced by *MISO's* expression will help define the exact mechanism mediating *Anopheles* tolerance to *P. falciparum* infection. Combined with the finding that *MISO* is highly induced by mating in *An. coluzzii* females collected from natural mating swarms in the same villages in Burkina Faso [43], our results reveal that this female protein has an important function for maintaining reproductive fitness of field malaria vectors during *P. falciparum* infection.

Materials and methods

Ethics statement

Procedures involving human subjects were conducted in accordance with protocols approved by the Institutional Ethics Committee of the Centre Muraz (A003-2012/CE-CM), Bobo-Dioulasso, Burkina Faso and the Ethics Committee of the General Administration Department of the Ministry of Health of Burkina Faso (2014–0040). Volunteers with gametocytemias above 70 gametocytes/µl of blood were enrolled for the study following parental consent. *Anopheles* female mosquitoes at the Institut de Recherche en Sciences de la Santé (IRSS)/Centre Muraz, Bobo-Dioulasso, Burkina Faso were fed on rabbit blood in accordance with protocols approved by the Office of Laboratory Animal Welfare of the U.S. Public Health Service (Assurance Number: A5928-01) and the national committee of Burkina Faso (IRB registration #00004738 and FWA 00007038). Rabbits were cared for by veterinarians and trained personnel.

Rearing of Anopheles mosquitoes

The G3 strain of Anopheles gambiae (composed of the two closely related sibling species An. gambiae s.s. and Anopheles coluzzii, herein referred to as An. gambiae) and the SDA-500 strain of Anopheles stephensi were used in laboratory experiments at the Harvard T. H. Chan School of Public Health (HSPH). The identity of the G3 and SDA-500 strain was confirmed by PCR and DNA sequencing of resulting amplified product [44]. For experiments conducted at the IRSS in Burkina Faso, An. coluzzii mosquitoes were used from a colony established in 2008 from wild-caught gravid females collected in the vicinity of Bobo-Dioulasso, repeatedly replenished with F1 from wild-caught mosquito females collected in Kou Valley (11°23'14"N, 4° 24'42"W), 30 km from Bobo-Dioulasso [45,46], and identified by routine PCR-RFLP [44]. All mosquitoes were reared under standard conditions (26-28°C, 65-80% relative humidity, 12-hour:12-hour light:darkness photoperiod). Adults from laboratory mosquito strains were fed on 10% glucose solution supplied ad libitum, and females were fed on human blood weekly (Research Blood Components, Boston, MA). Adults from the IRSS mosquito strain were fed on 5% glucose solution supplied ad libitum, and females were fed on rabbit blood weekly. For laboratory and field experiments, to ensure virgin status of mosquitoes prior to mating experiments, male and female mosquitoes were separated as pupae.

Culturing Plasmodium falciparum NF54 parasites

The NF54 strain was confirmed as *P. falciparum* by PCR and DNA sequencing of amplified product [47]. Cultures were maintained at 37°C in an incubator gassed to 5% oxygen, 5% carbon dioxide, 90% nitrogen. Asexual cultures of NF54 were maintained below 2% parasitemia at 5% hematocrit (O+ red blood cells, RBC) (Interstate Blood Bank) in complete medium composed of RPMI 1640 with L-glutamine and 25 mM HEPES, 10 µg/ml hypoxanthine, 0.2%

sodium bicarbonate, and 10% heat-inactivated O+ human serum (Interstate Blood Bank) according to established protocols [48]. To induce gametocytogenesis, cultures were grown to parasitemias of 3–10%, then split to 2% parasitemia in complete media at 5% hematocrit [49]. Parasites were cultured without subsequent splitting for 14–20 days to generate stage IV and V gametocytes, changing the media daily.

Collection of natural isolates of Plasmodium falciparum parasites

Surveys for *P. falciparum* gametocyte carriers were conducted in the villages surrounding Bobo-Dioulasso, Burkina Faso. Blood smears were collected from pinprick samples from children ages 5–13, stained with 10% Giemsa, and assessed for *P. falciparum* infection levels by microscopy. The number of asexual and sexual parasites present per 1000 white blood cells was counted, and translated to gametocytes per μ l of blood using the conversion 8000 white blood cells/ μ l of blood.

Mating and fecundity assays in Anopheles mosquitoes

In laboratory assays. For laboratory experiments with An. gambiae and An. stephensi, matings were performed via two techniques: 1) 4 day-old virgin females were progressively added to cages containing 4-5 day-old males at a 1:2 ratio. Couples were then captured in cop*ula* as they dropped to the bottom of the cage using modified 50 mL centrifuge tubes [50]. 2) Females were mated using the forced mating technique, where the abdomen of anesthetized females are brought into contact with the abdomens of beheaded males to stimulate a mating response (https://www.beiresources.org). Females that failed to mate were screened out from the experiment by checking the abdomens of anesthetized females on a stereomicroscope Leica M80 with Fluorescence Light Source EL 6000 for autofluorescence resulting from the presence of a mating plug in the female atrium. All virgin females were separated from males as pupae and not subsequently exposed to males. Females were fed on P. falciparum-infectious or non-infectious blood at 1 day (d) post-mating (pM). 48 hours h pIBF, females were collected and stored in 70% ethanol at 4°C. This timepoint was selected because starting at 3 d pIBF, females lay their eggs on the floor of the cage despite being deprived of a proper oviposition site due to the humidity in the custom-built glove box utilized in these experiments (Inert Technology, Amesbury, MA; see detailed infection protocol below). Of note, this experimental set-up prevents us from collecting paired oocyst intensity and egg number in the same mated female. Fecundity was assessed by dissecting the ovaries of these females in 1X phosphate-buffered saline (1X PBS) and manually counting eggs developed on a Nikon SMZ1000 stereomicroscope. A subset of females was kept to 7 d pIBF to assess P. falciparum infection intensity and prevalence.

In field assays. Two days post-eclosion, *An. coluzzii* females were added to cages of 3– 4-day-old males in a 1:2 ratio. Following exposure to males over two nights, females were fed on *P. falciparum*-infected blood and transferred to new cages without males. Females were subsequently deprived of oviposition sites, and 7 d pIBF, females were collected and dissected to count the number of eggs developed and the number of oocysts in the midgut. Mating status was determined by assessing the presence/absence of sperm in the spermatheca. Cages were assessed regularly throughout the experiment to verify that females did not oviposit their eggs over the 7 day incubation period; no eggs were detected in the cages.

Infections of Anopheles mosquitoes with P. falciparum

Infections with *P. falciparum* **NF54.** Cages of mosquitoes were fed on gametocyte cultures spun down and diluted in fresh RBC and human serum (Interstate Blood Bank) in a

custom-built glove box (Inert Technology, Amesbury, MA). To heat-inactivate gametocytes as non-infectious controls, aliquots of resuspended cultures were incubated at 42°C shaking at 500 rpm for 15 minutes [51]. 160 μ l of these infectious and non-infectious culture samples were loaded into custom-designed glass membrane feeders (Chemglass Life Sciences) sealed with parafilm and heated to 37°C with a Haake D1 Immersion Circulator. Mosquitoes were allowed to feed for 30–60 minutes, adding more culture as necessary. All non- and partially-fed females were removed, and fully blood-fed mosquitoes were fed with 10% glucose solution *ad libitum* until dissection.

Infections with natural *P. falciparum* **isolates.** 2–4 ml of venous blood was drawn from gametocyte carriers, spun down, and serum was removed and replaced with an equal volume of *Plasmodium*-naïve AB serum. The sample was resuspended and loaded into custom-designed glass membrane feeders sealed with parafilm and heated to 37°C with a Julabo ED Immersion Circulator. Mosquitoes in cups were allowed to feed for 60–90 minutes. All non-and partially-fed females were removed from cages following the infection, and fully-fed mosquitoes were supplied 5% glucose solution *ad libitum* until dissection.

Oocyst counts. 7 d pIBF, female mosquitoes were collected in ethanol and transferred to 1X PBS. Midguts were dissected in 1X PBS, stained in 2 mg/ml mercury dibromofluorescein disodium salt (mercurochrome) (Sigma-Aldrich, St. Louis, MO) for 15 minutes, then loaded onto slides for visualization and manual counting at 200X using an Olympus Inverted CKX41 microscope.

Gene expression knock-down using dsRNA

Generating dsRNA from plasmids. The plasmids utilized in these experiments were constructed for previous publications: pLL10-MISO (containing a 397 bp fragment of *MISO*, *AGAP002620*) [28] and pLL10-LacZ (containing a 816 bp fragment of *LacZ*) [28,52]. Plasmids were harvested from NEB Turbo chemically-competent *E. coli* cells (New England Biolabs, Ipswich, MA) and verified by DNA sequencing. Using primers containing the T7 sequence and specific to the plasmid backbone, PCR reactions were performed on plasmid DNA to amplify the coding sequence and generate knock-down constructs specific to each plasmid. The primers used to amplify the *MISO* and *LacZ* cassettes were pLL10-FWD: 5'-<u>TAATACGACTCACC TATAGGGCTCGAGGTCGACGGTATCG-3' and pLL10-REV: 5'-<u>TAATACGACTCACCTAGGGCCGCCTCTAGAACTAG-3'</u> (T7 regions underlined). PCR product sizes were confirmed via gel electrophoresis. dsRNA was then generated by incubating each PCR product with T7 enzyme using the MEGAscript T7 Transcription Kit (Thermo Fisher Scientific), purifying the dsRNA product by phenol-chloroform extraction, and re-suspending purified dsRNA in molecular grade de-ionized water (dH₂O) at a concentration of 10 µg/µl.</u>

dsRNA injections. 1-day-old females were randomly assigned to injection groups (ds*MISO* or ds*Control*), anesthetized, and injected with 69 nl of 10 μ g/ μ l dsRNA using a Drummond Scientific Nanoject II or III. Injected females were then mated and blood-fed on *P. falciparum*-infected blood as described above.

RNA extraction, cDNA synthesis, and qRT-PCR

To assess silencing efficiency in *MISO* RNAi experiments, 8–12 atria were dissected from ds*MISO* and control females in 1XPBS approximately 24 h pM, pooled in TRIzol Reagent (Thermo Fisher Scientific), and stored at –80°C. Samples were homogenized with a hand-held homogenizer on ice, and RNA was extracted in TRIzol Reagent using 1-bromo-3-chloropropane, precipitated in isopropanol, and washed in 75% ethanol. RNA was quantified using a Nanodrop 2000c, and equal quantities of atrial RNA from *MISO* knock-down and control

groups were treated with TURBO DNase. The DNase-treated RNA samples were then loaded into 100 μ l cDNA synthesis reactions with reagents at the following final concentrations: 1X First Strand Buffer, 5 mM DTT, 500 μ M dNTP mix, 0.4 units RNaseOUT recombinant ribonuclease inhibitor, 2.5 μ M random hexamers, and 1.25 units M-MLV reverse transcriptase (all reagents from Thermo Fisher Scientific). Primer sets (Integrated DNA Technologies) suitable for qRT-PCR were selected from a previous publication [28] where they were designed using Primer-BLAST (NCBI) [53] to limit off-target binding and to span exon-exon junctions when possible. The primer sequences were:

qRT-PCR MISO-F: AGACGATGGAGGGACTGATG

qRT-PCR MISO-R: GGATTCGCTTTCGTGCTG

qRT-PCR RPL19-F: CCAACTCGCGACAAAACATTC

qRT-PCR RPL19-R: ACCGGCTTCTTGATGATCAGA

Quantification was done using SYBR Green Detection, mixing 7.5 μ l 1X Fast SYBR Green Master Mix (Thermo Fisher Scientific), 5 μ l of a 1:10 dilution of cDNA, and primers at optimal concentrations, in duplicate for each sample on a StepOnePlus Real-Time PCR System (Thermo Fisher Scientific). Relative quantification of gene expression was calculated using the $\Delta C_{\rm T}$ method, with the ribosomal protein *RpL19 (AGAP004422)* used as the housekeeping gene. Samples were excluded from the analysis if the presence of >1 melt curve peak was detected or if amplification occurred in the non-template controls. Mean *MISO* silencing efficiency in *dsMISO*-injected females was 61% across 4 experiments.

Quantification and statistical analysis

All statistical analyses were performed in R (version 4.0.2) [54]. A summary of all model statistics can be found in <u>S1 Table</u>. We used the glmmTMB package [55] to build Generalized Mixed Linear Models (GLMMs). For each of the models, we chose the most appropriate distribution through Akaike information criteria (AIC) comparisons. GLMMs with Gaussian errors were used to test the effects of mating, infection, and their interaction on the number of developing eggs in the ovaries of An. gambiae (Fig 1A) and An. stephensi (Fig 1B). A GLMM with binomial errors was used to test the effect of mating, infection, and their interaction on the likelihood of developing eggs (egg prevalence) in An. stephensi (Fig 1B). Similar binomial GLMMs were used to test the effect of mating on oocyst prevalence in An. gambiae (Fig 2A) and An. stephensi (Fig 2B). GLMMs with zero-truncated negative binomial errors were used to test the effect of mating on oocyst intensity (the number of oocysts in infected mosquito females) in both An. gambiae (Fig 2A) and An. stephensi (Fig 2B). For each of these GLMMs, experimental replicate was set as a random effect. The effects of gametocytemia, MISO silencing (referred to as "treatment") and their interaction on egg number in An. coluzzii (Fig 3A) was analyzed using a GLMM with zero-inflated negative binomial errors. The effects of gametocytemia, treatment, and their interaction on oocyst prevalence and intensity in An. coluzzii were analyzed using a binomial GLMM and a zero-truncated negative binomial GLMM, respectively (Fig 3B). The effects of oocyst intensity, treatment, and their interaction on egg number and prevalence in An. coluzzii were analyzed using a zero-inflated negative binomial GLMM (Fig 3C) and a binomial GLMM (Fig 3D), respectively. In these GLMMs, because the experiment on An. coluzzii was conducted over 3 replicates and using a total of 5 gametocyte carriers (one replicate with one carrier and two replicates with two carriers each), carrier (parasite isolate) was nested within replicate and considered together as nested random effects. Model simplification used stepwise removal of terms, followed by likelihood ratio tests (LRT). Term removals that significantly reduced explanatory power (p < 0.05) were retained in the minimal adequate model. All pairwise comparisons were done using the "emmeans" function

in the emmeans R package that computes estimated marginal means and automatically corrects for unequal sample sizes using the Tukey-Kramer method [56].

Supporting information

S1 Table. Model statistics related to analyses in Figs 1–3. Generalized Linear Mixed Models (GLMMs) were constructed and analyzed in R. The test/model used in each figure and for each comparison is outlined with individual likelihood ratio test (LRT) outputs. Significant tests (p < 0.05) are bolded. (PDF)

S2 Table. Parameters for experimental infections of *An. coluzzii* with field isolates of *P. falciparum*. Samples of *P. falciparum*-infected blood were collected from 5 gametocyte donors and fed to *An. coluzzii* mosquitoes silenced for *MISO* or a control gene. The number of gametocytes for each sample was counted per 1000 white blood cells on blood smears, then converted to the number of gametocytes per μ l of blood assuming a standard white blood cell count of 8000 cells per μ l of blood. The outcome of infection was determined by counting the number of oocysts developed per female 7 d pIBF. (PDF)

Acknowledgments

We wish to thank Kate Thornburg, Emily Lund, and David Clarke for help with mosquito rearing and insectary procedures. We also wish to thank Naresh Singh, Jamaica Siwak, and Ping Lui for assistance with *P. falciparum* culture and infections in the laboratory at the Harvard T. H. Chan School of Public Health, and Bali-Jean Bazié and Raymond Hien for assistance with *P. falciparum* infections and mosquito dissections at the Institut de Recherche en Sciences de la Santé in Bobo-Dioulasso, Burkina Faso. We are grateful to Daniel E. Neafsey (Harvard T.H. Chan School of Public Health, Boston, MA, USA) for helpful discussions and feedback on this manuscript. We would also like to thank Carolina Barillas-Mury and Alvaro-Molina Cruz (National Institutes of Health, Bethesda, MD, USA) for guidance and resource sharing for *P. falciparum* NF54 culturing.

Author Contributions

Conceptualization: Perrine Marcenac, Flaminia Catteruccia.

Formal analysis: Perrine Marcenac, Adam South, Thierry Lefèvre.

Funding acquisition: Perrine Marcenac, Flaminia Catteruccia.

Investigation: Perrine Marcenac, W. Robert Shaw, Evdoxia G. Kakani, Sara N. Mitchell, Eryney Marrogi, Daniel G. Abernathy, Rakiswendé Serge Yerbanga, Abdoulaye Diabaté, Thierry Lefèvre.

Methodology: Perrine Marcenac, Flaminia Catteruccia.

Resources: Roch K. Dabiré.

Supervision: Thierry Lefèvre, Flaminia Catteruccia.

Visualization: Perrine Marcenac, Thierry Lefèvre.

Writing - original draft: Perrine Marcenac, Flaminia Catteruccia.

Writing – review & editing: Perrine Marcenac, W. Robert Shaw, Evdoxia G. Kakani, Sara N. Mitchell, Kristine Werling, Thierry Lefèvre, Flaminia Catteruccia.

References

- 1. WHO. World Malaria Report. World Health Organization; 2018.
- Moreno M, Marinotti O, Krzywinski J, Tadei WP, James AA, Achee NL, et al. Complete mtDNA genomes of Anopheles darlingi and an approach to anopheline divergence time. Malar J. 2010; 9: 127. https://doi.org/10.1186/1475-2875-9-127 PMID: 20470395
- Neafsey DE, Waterhouse RM, Abai MR, Aganezov SS, Alekseyev MA, Allen JE, et al. Mosquito genomics. Highly evolvable malaria vectors: the genomes of 16 Anopheles mosquitoes. Science. 2015; 347 (6217): 1258522. https://doi.org/10.1126/science.1258522 PMID: 25554792
- Sinka ME, Bangs MJ, Manguin S, Rubio-Palis Y, Chareonviriyaphap T, Coetzee M, et al. A global map of dominant malaria vectors. Parasit Vectors. 2012; 5: 69. <u>https://doi.org/10.1186/1756-3305-5-69</u> PMID: 22475528
- Liu W, Li Y, Learn GH, Rudicell RS, Robertson JD, Keele BF, et al. Origin of the human malaria parasite Plasmodium falciparum in gorillas. Nature. 2010; 467(7314): 420–425. <u>https://doi.org/10.1038/nature09442</u> PMID: 20864995
- Sundararaman SA, Plenderleith LJ, Liu W, Loy DE, Learn GH, Li Y, et al. Genomes of cryptic chimpanzee Plasmodium species reveal key evolutionary events leading to human malaria. Nat Commun. 2016; 7: 11078. https://doi.org/10.1038/ncomms11078 PMID: 27002652
- Makanga B, Yangari P, Rahola N, Rougeron V, Elguero E, Boundenga L, et al. Ape malaria transmission and potential for ape-to-human transfers in Africa. Proc Natl Acad Sci U S A. 2016; 113(19): 5329–5334. https://doi.org/10.1073/pnas.1603008113 PMID: 27071123
- Otto TD, Gilabert A, Crellen T, Bohme U, Arnathau C, Sanders M, et al. Genomes of all known members of a Plasmodium subgenus reveal paths to virulent human malaria. Nat Microbiol. 2018; 3(6): 687–697. https://doi.org/10.1038/s41564-018-0162-2 PMID: 29784978
- Loy DE, Liu W, Li Y, Learn GH, Plenderleith LJ, Sundararaman SA, et al. Out of Africa: origins and evolution of the human malaria parasites Plasmodium falciparum and Plasmodium vivax. Int J Parasitol. 2017; 47(2–3): 87–97. https://doi.org/10.1016/j.ijpara.2016.05.008 PMID: 27381764
- Ferguson HM, Read AF. Why is the effect of malaria parasites on mosquito survival still unresolved? Trends Parasitol. 2002; 18(6): 256–261. <u>https://doi.org/10.1016/s1471-4922(02)02281-x</u> PMID: 12036738
- Almeida AP, Billingsley PF. Induced immunity against the mosquito Anopheles stephensi (Diptera: Culicidae): effects of cell fraction antigens on survival, fecundity, and plasmodium berghei (Eucoccidiida: Plasmodiidae) transmission. J Med Entomol. 2002; 39(1): 207–214. <u>https://doi.org/10.1603/0022-2585-39.1.207</u> PMID: 11931258
- Jahan N, Hurd H. The effects of infection with Plasmodium yoelii nigeriensis on the reproductive fitness of Anopheles stephensi. Ann Trop Med Parasitol. 1997; 91(4): 365–369. <u>https://doi.org/10.1080/</u> 00034989760987 PMID: 9290843
- Ahmed AM, Maingon R, Romans P, Hurd H. Effects of malaria infection on vitellogenesis in Anopheles gambiae during two gonotrophic cycles. Insect Molecular Biology. 2001; 10(4): 347–356. <u>https://doi.org/10.1046/j.0962-1075.2001.00273.x PMID: 11520358</u>
- Ahmed AM, Maingon RD, Taylor PJ, Hurd H. The effects of infection with Plasmodium yoelii nigeriensis on the reproductive fitness of the mosquito Anopheles gambiae. Invertebrate Reproduction & Development. 1999; 36(1–3): 217–222.
- Ferguson HM, Rivero A, Read AF. The influence of malaria parasite genetic diversity and anaemia on mosquito feeding and fecundity. Parasitology. 2003; 127(Pt 1): 9–19. <u>https://doi.org/10.1017/</u> s0031182003003287 PMID: 12885184
- Hogg JC, Hurd H. Plasmodium yoelii nigeriensis: the effect of high and low intensity of infection upon the egg production and bloodmeal size of Anopheles stephensi during three gonotrophic cycles. Parasitology. 1995; 111 (Pt 5): 555–562. https://doi.org/10.1017/s0031182000077027 PMID: 8559587
- Werling K, Shaw WR, Itoe MA, Westervelt KA, Marcenac P, Paton DG, et al. Steroid Hormone Function Controls Non-competitive Plasmodium Development in Anopheles. Cell. 2019; 177(2): 315–325 e314. https://doi.org/10.1016/j.cell.2019.02.036 PMID: 30929905
- Attardo GM, Hansen IA, Raikhel AS. Nutritional regulation of vitellogenesis in mosquitoes: Implications for anautogeny. Insect Biochemistry and Molecular Biology. 2005; 35(7): 661–675. https://doi.org/10. 1016/j.ibmb.2005.02.013 PMID: 15894184

- Raikhel AS. Vitellogenesis in mosquitoes. Advances in Disease Vector Research. New York, NY: Springer; 1992. p. 1–39.
- Costa G, Gildenhard M, Eldering M, Lindquist RL, Hauser AE, Sauerwein R, et al. Non-competitive resource exploitation within mosquito shapes within-host malaria infectivity and virulence. Nat Commun. 2018; 9(1): 3474. https://doi.org/10.1038/s41467-018-05893-z PMID: 30150763
- Rono MK, Whitten MM, Oulad-Abdelghani M, Levashina EA, Marois E. The major yolk protein vitellogenin interferes with the anti-plasmodium response in the malaria mosquito Anopheles gambiae. PLoS Biol. 2010; 8(7): e1000434. https://doi.org/10.1371/journal.pbio.1000434 PMID: 20652016
- 22. McNamara KB, van Lieshout E, Simmons LW. Females suffer a reduction in the viability of stored sperm following an immune challenge. J Evol Biol. 2014; 27(1): 133–140. https://doi.org/10.1111/jeb. 12278 PMID: 24251540
- Baer B, Armitage SA, Boomsma JJ. Sperm storage induces an immunity cost in ants. Nature. 2006; 441(7095): 872–875. https://doi.org/10.1038/nature04698 PMID: 16778889
- Paynter E, Millar AH, Welch M, Baer-Imhoof B, Cao D, Baer B. Insights into the molecular basis of longterm storage and survival of sperm in the honeybee (Apis mellifera). Sci Rep. 2017; 7: 40236. https:// doi.org/10.1038/srep40236 PMID: 28091518
- 25. Yuval B. Mating systems of blood-feeding flies. Annu Rev Entomol. 2006; 51: 413–440. <u>https://doi.org/</u> 10.1146/annurev.ento.51.110104.151058 PMID: 16332218
- 26. Shaw WR, Teodori E, Mitchell SN, Baldini F, Gabrieli P, Rogers DW, et al. Mating activates the heme peroxidase HPX15 in the sperm storage organ to ensure fertility in Anopheles gambiae. Proceedings of the National Academy of Sciences. 2014; 111(16): 5854–5859. <u>https://doi.org/10.1073/pnas.</u> 1401715111 PMID: 24711401
- Gabrieli P, Kakani EG, Mitchell SN, Mameli E, Want EJ, Mariezcurrena Anton A, et al. Sexual transfer of the steroid hormone 20E induces the postmating switch in Anopheles gambiae. Proc Natl Acad Sci U S A. 2014; 111(46): 16353–16358. https://doi.org/10.1073/pnas.1410488111 PMID: 25368171
- Baldini F, Gabrieli P, South A, Valim C, Mancini F, Catteruccia F. The Interaction between a Sexually Transferred Steroid Hormone and a Female Protein Regulates Oogenesis in the Malaria Mosquito Anopheles gambiae. PLoS Biol. 2013; 11(10): e1001695. https://doi.org/10.1371/journal.pbio.1001695 PMID: 24204210
- Mitchell SN, Kakani EG, South A, Howell PI, Waterhouse RM, Catteruccia F. Mosquito biology. Evolution of sexual traits influencing vectorial capacity in anopheline mosquitoes. Science. 2015; 347(6225): 985–988. https://doi.org/10.1126/science.1259435 PMID: 25722409
- Pondeville E, Puchot N, Lang M, Cherrier F, Schaffner F, Dauphin-Villemant C, et al. Evolution of sexually-transferred steroids and mating-induced phenotypes in Anopheles mosquitoes. Sci Rep. 2019; 9 (1): 4669. https://doi.org/10.1038/s41598-019-41094-4 PMID: 30874601
- **31.** Roy BA, Kirchner JW. Evolutionary dynamics of pathogen resistance and tolerance. Evol. 2000; 54(1): 51–63. https://doi.org/10.1111/j.0014-3820.2000.tb00007.x PMID: 10937183
- Medzhitov R, Schneider DS, Soares MP. Disease tolerance as a defense strategy. Science. 2012; 335 (6071): 936–941. https://doi.org/10.1126/science.1214935 PMID: 22363001
- Ayres JS, Schneider DS. Tolerance of infections. Annu Rev Immunol. 2012; 30: 271–294. <u>https://doi.org/10.1146/annurev-immunol-020711-075030 PMID: 22224770</u>
- 34. Dahalan FA, Churcher TS, Windbichler N, Lawniczak MKN. The male mosquito contribution towards malaria transmission: Mating influences the Anopheles female midgut transcriptome and increases female susceptibility to human malaria parasites. PLoS Pathog. 2019; 15(11): e1008063. https://doi.org/10.1371/journal.ppat.1008063 PMID: 31697788
- Schwenke RA, Lazzaro BP, Wolfner MF. Reproduction-Immunity Trade-Offs in Insects. Annu Rev Entomol. 2016; 61: 239–256. https://doi.org/10.1146/annurev-ento-010715-023924 PMID: 26667271
- Molina-Cruz A, Garver LS, Alabaster A, Bangiolo L, Haile A, Winikor J, et al. The human malaria parasite Pfs47 gene mediates evasion of the mosquito immune system. Science. 2013; 340(6135): 984– 987. https://doi.org/10.1126/science.1235264 PMID: 23661646
- Molina-Cruz A, Canepa GE, Kamath N, Pavlovic NV, Mu J, Ramphul UN, et al. Plasmodium evasion of mosquito immunity and global malaria transmission: The lock-and-key theory. Proc Natl Acad Sci U S A. 2015; 112(49): 15178–15183. https://doi.org/10.1073/pnas.1520426112 PMID: 26598665
- Molina-Cruz A, Zilversmit MM, Neafsey DE, Hartl DL, Barillas-Mury C. Mosquito Vectors and the Globalization of Plasmodium falciparum Malaria. Annu Rev Genet. 2016; 50: 447–465. <u>https://doi.org/10. 1146/annurev-genet-120215-035211</u> PMID: 27732796
- Goic B, Stapleford KA, Frangeul L, Doucet AJ, Gausson V, Blanc H, et al. Virus-derived DNA drives mosquito vector tolerance to arboviral infection. Nat Commun. 2016; 7: 12410. <u>https://doi.org/10.1038/ncomms12410</u> PMID: 27580708

- Roesijadi G, Rezvankhah S, Binninger DM, Weissbach H. Ecdysone induction of MsrA protects against oxidative stress in Drosophila. Biochem Biophys Res Commun. 2007; 354(2): 511–516. https://doi.org/ 10.1016/j.bbrc.2007.01.005 PMID: 17239346
- Krishnan N, Vecera J, Kodrik D, Sehnal F. 20-Hydroxyecdysone prevents oxidative stress damage in adult Pyrrhocoris apterus. Arch Insect Biochem Physiol. 2007; 65(3): 114–124. <u>https://doi.org/10.1002/arch.20182 PMID: 17570141</u>
- Kodrik D, Bednarova A, Zemanova M, Krishnan N. Hormonal Regulation of Response to Oxidative Stress in Insects-An Update. Int J Mol Sci. 2015; 16(10): 25788–25816. <u>https://doi.org/10.3390/</u> ijms161025788 PMID: 26516847
- 43. Thailayil J, Gabrieli P, Caputo B, Bascunan P, South A, Diabate A, et al. Analysis of natural female post-mating responses of Anopheles gambiae and Anopheles coluzzii unravels similarities and differences in their reproductive ecology. Sci Rep. 2018; 8(1): 6594. <u>https://doi.org/10.1038/s41598-018-24923-w PMID: 29700344</u>
- Santolamazza F, Mancini E, Simard F, Qi Y, Tu Z, della Torre A. Insertion polymorphisms of SINE200 retrotransposons within speciation islands of Anopheles gambiae molecular forms. Malar J. 2008; 7: 163. https://doi.org/10.1186/1475-2875-7-163 PMID: 18724871
- 45. Harris C, Morlais I, Churcher TS, Awono-Ambene P, Gouagna LC, Dabire RK, et al. Plasmodium falciparum produce lower infection intensities in local versus foreign Anopheles gambiae populations. PLoS ONE. 2012; 7(1): e30849. https://doi.org/10.1371/journal.pone.0030849 PMID: 22292059
- Hien DF, Dabire KR, Roche B, Diabate A, Yerbanga RS, Cohuet A, et al. Plant-Mediated Effects on Mosquito Capacity to Transmit Human Malaria. PLoS Pathog. 2016; 12(8): e1005773. https://doi.org/ 10.1371/journal.ppat.1005773 PMID: 27490374
- Snounou G, Viriyakosol S, Zhu XP, Jarra W, Pinheiro L, do Rosario VE, et al. High sensitivity of detection of human malaria parasites by the use of nested polymerase chain reaction. Mol Biochem Parasitol. 1993; 61(2): 315–320. https://doi.org/10.1016/0166-6851(93)90077-b PMID: 8264734
- Trager W, Jensen JB. Human malaria parasites in continuous culture. Science. 1976; 193(4254): 673– 675. https://doi.org/10.1126/science.781840 PMID: 781840
- Ifediba T, Vanderberg JP. Complete in vitro maturation of Plasmodium falciparum gametocytes. Nature. 1981; 294(5839): 364–366. https://doi.org/10.1038/294364a0 PMID: 7031476
- Rogers DW, Whitten MM, Thailayil J, Soichot J, Levashina EA, Catteruccia F. Molecular and cellular components of the mating machinery in Anopheles gambiae females. Proc Natl Acad Sci U S A. 2008; 105(49): 19390–19395. https://doi.org/10.1073/pnas.0809723105 PMID: 19036921
- Mendes AM, Awono-Ambene PH, Nsango SE, Cohuet A, Fontenille D, Kafatos FC, et al. Infection intensity-dependent responses of Anopheles gambiae to the African malaria parasite Plasmodium falciparum. Infect Immun. 2011; 79(11): 4708–4715. <u>https://doi.org/10.1128/IAI.05647-11</u> PMID: 21844236
- 52. Blandin S, Shiao SH, Moita LF, Janse CJ, Waters AP, Kafatos FC, et al. Complement-like protein TEP1 is a determinant of vectorial capacity in the malaria vector Anopheles gambiae. Cell. 2004; 116(5): 661–670. https://doi.org/10.1016/s0092-8674(04)00173-4 PMID: 15006349
- Ye J, Coulouris G, Zaretskaya I, Cutcutache I, Rozen S, Madden TL. Primer-BLAST: a tool to design target-specific primers for polymerase chain reaction. BMC Bioinformatics. 2012; 13: 134. https://doi. org/10.1186/1471-2105-13-134 PMID: 22708584
- 54. Crawley MJ. The R Book. Chichester, United Kingdom: Wiley; 2012.
- 55. Brooks ME, Kristensen K, van Benthem KJ, Magnusson A, Berg CW, Nielsen A, et al. glmmTMB Balances Speed and Flexibility Among Packages for Zero-inflated Generalized Linear Mixed Modeling. The R Journal. 2017; 9(2). PMID: 29564164
- 56. Lenth R. emmeans: Estimated Marginal Means, aka Least-Squares Means. R package version 1.4.8, 2020.